Author's response to reviews

Title: Severe congenital neutropenia caused by the ELANE gene mutation in a Vietnamese boy with misdiagnosis of tuberculosis and autoimmune neutropenia: a case report

Authors:

Quang Van Vu (vvquang@hpmu.edu.vn)
Taizo Wada (taizo@staff.kanazawa-u.ac.jp)
Tham Thi Tran (tttham@hpmu.edu.vn)
Duc Ngoc Ngo (ngocducbsnt@gmail.com)
Thuc Van Dinh (dvthuc@hpmu.edu.vn)
Cuong Hung Nguyen (cuongshpthp@yahoo.com.vn)
Huong Thi Minh Le (lehuong@mail.ru)
Akihiro Yachie (yachie@staff.kanazawa-u.ac.jp)
Sang Ngoc Nguyen (nnsang@hpmu.edu.vn)

Version: 2
Date: 29 July 2014

Author's response to reviews: see over
Dear Editors,

Please find a file of “Severe congenital neutropenia caused by the ELANE gene mutation in a Vietnamese boy with misdiagnosis of tuberculosis and autoimmune neutropenia: a case report” by Quang Van Vu, Taizo Wada, Tham Thi Tran, Duc Ngoc Ngo, Thuc Van Dinh, Cuong Hung Nguyen, Huong Thi Minh Le, Akihiro Yachie, and Sang Ngoc Nguyen. We are submitting this paper to be considered for publication as a research article in your journal. No part of this paper has been published or accepted by any other journal.

We believe that your readers will find this paper interesting because our findings could tell physicians that immunodeficiency diseases are hidden in large numbers of children with "common" infections.

Please direct all correspondence about this paper to me following address:

Quang Van Vu, MD, PhD
Department of Pediatrics,
Haiphong Medical University, Vietnam
72 A Nguyen Binh Khiem, Ngo Quyen, Haiphong, Vietnam
Phone: +84-313-733-311
Fax: +84-313-733-315
E-mail: vvquang@hpmu.edu.vn

Thank you for considering our paper. We look forward to receiving your reply.

Sincerely,